Author's response to reviews

Title: Evaluation of the Kinase Domain of c-KIT in Canine Cutaneous Mast Cell Tumors

Authors:

Joshua D Webster (webste28@cvm.msu.edu)
Matti Kiupel (kiupel@dcpah.msu.edu)
Vilma Yuzbasiyan-Gurkan (yuzbasiyan@cvm.msu.edu)

Version: 5 Date: 22 March 2006

Author's response to reviews:

It appears that the text originally used in figure 4 is not being recognized. I have changed the font for this figure and am uploading a new version of this figure.
Thank you,
Joshua Webster